Author's response to reviews

Title: Quantitative pedigree analysis and mitochondrial DNA sequence variants in adults with cyclic vomiting syndrome.

Authors:

Thangam Venkatesan (tvenkate@mcw.edu)
Nilay Kumar (nikumar@mcw.edu)
Jyotirmoy Sengupta (rsengupta@nrconsults.com)
Muhammad Ali (muhammad.ali@phhs.org)
Baber Malik (babermmalik@gmail.com)
Aniko Szabo (aszabo@mcw.edu)
Miranda A.L. van Tilburg (tilburg@med.unc.edu)
Essam A. Zaki (ezaki@chla.usc.edu)
Richard G. Boles (rboles@chla.usc.edu)

Version: 4 Date: 23 September 2014

Author's response to reviews: see over
Dear Dr. Morawska,

We have made the additional corrections as requested by the editorial team.

Please see below:

1) Please format the Authors Contribution section into a paragraph.
   Done

2) Please include the funding information in your acknowledgements section and not as a separate section.
   Done

3) Please remove the heading from figure one and to include this in the manuscript instead.
   Done

4) Please provide a list of additional files.
   Done

Thank you

Sincerely,

Thangam